Author’s response to reviews

Title: Actions of sesquiterpene lactones isolated from Moquiniastrum polymorphum subsp. floccosum in MCF7 cell line and their potentiating action on doxorubicin

Authors:

Mariana Mauro (mari_mauro84@hotmail.com)
Renata Matuo (matuorenata@yahoo.com.br)
Natan de David (natan.dedavid@gmail.com)
Regiane Strapasson (regi_0112@hotmail.com)
Rodrigo Juliano Oliveira (rodrigo.oliveira@ufms.br)
Maria Élida Stefanello (elida@ufpr.br)
Cândida Aparecida Kassuya (candida2005@gmail.com)
Maria de Fátima Matos (matosmfc@gmail.com)
Fábio José Faria (fabio.faria@ufms.br)
Deiler Costa (deilercosta@yahoo.com.br)

Version: 3 Date: 12 Jun 2017

Author’s response to reviews:

Dear Editorial Team;

We appreciate the notes relating to the authors. We have made the requested changes. Below is further clarification of changes:

Reviewers: We notice one of the listed authors (Strapasson) isn't included on the title page. Please ensure all authors are provided on the title page in the proper author order. We also notice the order of authors provided on the title page is different from the order listed in the submission data. Please confirm the correct order of authorship and provide the correct order in both the submission system and on the title page.

Authors: The missing author (Strapasson) was added to the title page and the order of authors was stowed in accordance with the order that is present in the submission data.
Best wishes;